Author’s response to reviews

Title: Surgical treatment of gastric venous congestion in association with extended resection of pancreas

Authors:
Shuji Kagota (sur165@osaka-med.ac.jp)
Tetsunosuke Shimizu (0315shimizu@gmail.com)
Kohei Taniguchi (sur144@osaka-med.ac.jp)
Atsushi Tomioka (sur168@osaka-med.ac.jp)
Yoshihiro Inoue (sur129@osaka-med.ac.jp)
Koji Komeda (sur115@osaka-med.ac.jp)
Mitsuhiro Asakuma (sur105@osaka-med.ac.jp)
Sang-Woong Lee (sur079@osaka-med.ac.jp)
Fumitoshi Hirokawa (sur122@osaka-med.ac.jp)
Kazuhisa Uchiyama (uchi@osaka-med.ac.jp)

Version: 2 Date: 31 Jan 2020

Author’s response to reviews:

RE: BSUR-D-19-00878

Surgical treatment of gastric venous congestion in association with extended resection of pancreas

Dear Assistant Editor

BMC Surgery

Thank you very much for your letter dated on January 28th, 2020.
We found the your kind and detailed suggestions most helpful, and have revised the manuscript accordingly.

Please find the revised manuscript. We resubmit this revised manuscript for your consideration for publication in BMC Surgery.

We are sincerely looking forward to hearing from you again in your next earliest convenience.

Sincerely,

Tetsunosuke Shimizu, M.D.
Department of General and Gastroenterological Surgery, Osaka Medical College Hospital
Email: sur116@poh.osaka-med.ac.jp
Address: 2-7, Daigaku-cho, Takatsuki City, Osaka 569-8686, Japan.
Phone: +81 072 (683) 1221
Fax: +81 72 (685) 2057

Our responses to the referees' reports are as follows:

To EDITORIAL COMMENTS:

1. Authors Contributions

Thank you for pointing out the important things. We rewrote.

2. Consent for Publication

Thank you for your comments. I already wrote.
3. Cite
Thank you for your comments. I cited all figures.

4. Clean Manuscript
I rewrote our manuscript, and uploaded.